Author’s response to reviews

Title: Streptococcal toxic shock syndrome caused by the dissemination of an invasive emm3/ST15 strain of Streptococcus pyogenes

Authors:
Tsuyoshi Sekizuka (sekizuka@niid.go.jp)
Emina Nai (naiemina19870112@gmail.com)
Tomohiro Yoshida (tomoyoshida0104@gmail.com)
Shota Endo (shte92@gmail.com)
Emi Hamajima (emi.tam988@gmail.com)
Satoka Akiyama (satoka0919@gmail.com)
Yoji Ikuta (y-ikuta@umin.ac.jp)
Natsuko Obana (na2co3_716@yahoo.co.jp)
Takahiro Kawaguchi (kawaguchi0421@gmail.com)
Kenta Hayashi (kentahayashi712@gmail.com)
Masahiro Noda (masa.noda5279@gmail.com)
Tomoko Sumita (ktomoko0826@yahoo.co.jp)
Masayuki Kokaji (kokaji13272@gmail.com)
Tatsuo Katori (tkatori@ac.auone-net.jp)
Masanori Hashino (hashino@niid.go.jp)
Kunihiro Oba (kunihiro@zb3.so-net.ne.jp)
Makoto Kuroda (makokuro@niid.go.jp)

Version: 3 Date: 28 Nov 2017

Author’s response to reviews:
November 28, 2017

Edward Spofford,

BMC Infectious Diseases case report,

RE3: INFD-D-17-01180

Streptococcal toxic shock syndrome caused by the global dissemination of an invasive emm3/ST15 strain of Streptococcus pyogenes

Dear Edward Spofford,

We wish to resubmit our manuscript entitled “Streptococcal toxic shock syndrome caused by the global dissemination of an invasive emm3/ST15 strain of Streptococcus pyogenes” (RE3: INFD-D-17-01180) for publication as a Case Report in BMC Infectious Diseases.

As the editor suggested, the manuscript had been revised in accordance with the suggestions, and the uploaded as the final and clean version without any track changes.

I appreciate your valuable suggestions for our manuscript, again.

Sincerely,

Makoto Kuroda, PhD

Director, Pathogen Genomics Center, National Institute of Infectious Diseases

1-23-1 Toyama, Shinjuku-ku, Tokyo 162-8640, Japan

Phone: +81-3-5285-1111 ext. 2739

Fax: +81-3-5285-1166

E-mail: makokuro@niid.go.jp